\begin{table}
%\begin{table}%t1
\caption{\label{tab:GalProperties1}Basic properties of the sample galaxies.}
\par
%\centerline
{
\begin{tabular}{cccccccccccccccccc}
\hline
\hline
ID & Galaxy  & RA(2000) & Dec  (2000)  & Type\hspace{1mm} & Redshift  & $\sigma$         & $B_{\rm T}$ & $H_{\rm T}$ & $K_{\rm T}$ & \multicolumn{1}{c}{Age}   & Mg$_2$ & Mg$_2$ & \Hb     & Fe5335  & [$Z$/H] & [Fe/H]  & CaT     \\
   &                    &                        &                         &                  &                      & (km s$^{-1}$)    & (mag) & (mag) & (mag) & \multicolumn{1}{c}{(Gyr)} & (mag)  & (mag)  & (\AA) & (\AA) & (\AA) & (\AA) & (\AA) \\
(1)& (2)                & (3)                    & (4)                     & (5)              & (6)                  & (7)              & (8)   & (9)   & (10)  & \multicolumn{1}{c}{(11)}  & (12a)  & (12b)  & (13)    & (14)    & (15)    & (16)    & (17)    \\
\hline
\multicolumn{18}{c}{Galaxies with low-resolution spectra}\\
1 ~ &  \object{IC~4296}	 & 13 36 38.85  &$-$33 57 59.3  & E         & 0.0124(a)  & 337.0(a)  & 11.61  & 14.56  & 14.30 &   5.2(j)  &  0.323  & 0.340(a)& 1.22(m)  & 3.03(m)  & 0.39(j) & \textellipsis & \textellipsis  \\
2 ~ & \object{NGC~3641}	 & 11 21 08.85  &  +03 11 40.2  & E~pec    & 0.0058(b)  & 163.2(i)  & 14.10  & 15.92  & 15.17 &   6.6(k)  &  0.284  & 0.273(k)& 0.72(k)  & 2.39(k)  & \textellipsis & \textellipsis & 6.343  \\
3 ~ & \object{NGC~3818}	 & 11 41 57.50  &$-$06 09 20.0  & E5        & 0.0055(a)  & 187.5(a)  & 12.67  & 14.69  & 14.54 &   6.4(k)  &  0.315  & 0.322(a)& 1.67(k)  & 2.87(k)  & 0.37(o) &   0.16(o) & 6.242  \\
4 ~ & \object{NGC~4281}	 & 12 20 21.52  &  +05 23 12.4  &S0$^+$:~sp& 0.0089(c)  & 230.5(h)  & 12.25  & 13.91  & 13.80 & \textellipsis &  0.314  & \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis \\
5 ~ & \object{NGC~4472}	 & 12 29 46.76  &  +07 59 59.9  & E2        & 0.0033(a)  & 309.7(a)  &  9.37  & 15.34  & 15.01 &  10.0(k)  &  0.306  & 0.331(a)& 1.37(k)  & 3.11(k)  & 0.26(o) &   0.06(o) & 6.119  \\
6 ~ & \object{NGC~4478}	 & 12 30 17.53  &  +12 19 40.3  & E2        & 0.0047(a)  & 153.0(a)  & 12.36  & 15.62  & 15.45 &   6.9(k)  &  0.253  & 0.260(a)& 1.67(k)  & 2.81(k)  & 0.30(o) &   0.16(o) & 6.810  \\
7 ~ & \object{NGC~4564}	 & 12 36 27.01  &  +11 26 18.8  & E         & 0.0038(a)  & 171.1(a)  & 12.05  & 14.74  & 14.66 &   8.3(k)  &  0.321  & 0.329(a)& 1.54(k)  & 2.81(k)  & \textellipsis &  \textellipsis & 6.325  \\
8 ~ & \object{NGC~4594}	 & 12 39 59.43  &$-$11 37 22.9  & Sa(s) sp  & 0.0034(a)  & 259.1(a)  &  8.98  & 12.22  & 12.22 &   9.5(k)  &  0.330  & 0.338(a)& 1.44(k)  & \textellipsis & \textellipsis &   0.39(o) & 6.361  \\
9 ~ & \object{NGC~4621}	 & 12 42 02.39  &  +11 38 45.1  & E5        & 0.0014(a)  & 232.0(a)  & 10.57  & 10.28  & 10.16 &   9.1(k)  &  0.328  & 0.345(a)& 1.39(k)  & 2.98(k)  & 0.07(o) & \textellipsis & 6.055  \\
10~ & \object{NGC~4649}	 & 12 43 40.19  &  +11 33 08.9  & E2        & 0.0036(a)  & 368.5(a)  &  9.81  & 14.61  & 14.35 &  11.9(m)  &  0.338  & 0.347(a)& 1.29(k)  & 2.57(k)  & 0.29(o) &   0.06(o) & \textellipsis \\
11~ & \object{NGC~4697}	 & 12 48 35.70  &$-$05 48 03.0  & E6        & 0.0041(a)  & 171.7(a)  & 10.14  & 13.35  & 13.31 &   6.4(k)  &  0.297  & 0.296(a)& 1.69(k)  & 1.94(k)  & 0.06(o) &$-$0.03(o) & 6.185  \\
12~ & \object{NGC~5077}	 & 13 19 31.66  &$-$12 39 25.0  & E3-4      & 0.0090(d)  & 255.9(i)  & 12.38  & 16.21  & 14.93 &  15.0(j)  &  0.295  & 0.324(j)& 1.83(m)  & 2.64(m)  & 0.12(j) & \textellipsis & \textellipsis \\
13~ & \object{NGC~5576}	 & 14 21 03.60  &  +03 16 14.4  & E3        & 0.0050(e)  & 183.1(i)  & 11.85  & 14.94  & 14.74 & \textellipsis &  0.253  & 0.243(f)& 1.52(n)  & 3.11(l)  & \textellipsis &   0.02(o) & \textellipsis \\
14~ & \object{NGC~6909}	 & 20 27 38.60  &$-$47 01 34.0  & E+:       & 0.0090(f)  & 128.7(f)  & 12.61  & 14.25  & 14.32 & \textellipsis &  0.208  & 0.214(f)& 2.17(p)  & 2.32(p)  & \textellipsis & \textellipsis & \textellipsis \\
\multicolumn{18}{c}{}\\
\multicolumn{18}{c}{Galaxies with high-resolution spectra}\\
15~ & \object{Mrk~1055}         & 02 48 18.55  &$-$08 57 37.4  & S?           & 0.0366(v)  & \textellipsis & \textellipsis & 14.69  & 14.30 & \textellipsis & \textellipsis &  \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis \\
16~ & \object{NGC~1144}         & 02 55 12.32  &$-$00 11 01.7  & E            & 0.0288(q)  & \textellipsis & 13.78  & 15.18  & 15.06 & \textellipsis & \textellipsis & \textellipsis &  \textellipsis & \textellipsis &  \textellipsis & \textellipsis & \textellipsis \\
17~ & \object{NGC~1362}         & 03 33 53.08  &$-$20 16 57.7  & S0$^o$:~pec & 0.0041(t)  &  91(s) & \textellipsis & 16.11  & 15.68 & \textellipsis & \textellipsis & \textellipsis & 0.223(f) &  \textellipsis & \textellipsis & \textellipsis & \textellipsis \\
18~ & \object{NGC~4472}         & \multicolumn{16}{c}{see above}\\	            		     	                
19~ & \object{NGC~7714}         & 23 36 14.28  &  +02 09 17.8  & Sb(s):~pec  & 0.0093(p)  &  62(u) & 13.00  & 15.85  & 16.08 & \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis & \textellipsis & 4.7(u)\\
\hline
\end{tabular}}
\par
\medskip
Columns: (1) Identification number, used in the plots, (2) Galaxy
name, (3)--(4) coordinates, (5) morphological type from
\citet[][hereafter RC3]{dev91}, except for \object{Mrk~1055} and
\object{NGC~1144} from the Lyon Extragalactic Database (LEDA), (6)
redshift, (7) central velocity dispersion, (8) total observed blue
magnitude from RC3, (9)--(10) total observed $H$ and $K$ magnitude from
\citet[][2MASS All-Sky Catalog of point sources]{cut03}, (11) age,
(12)--(17) optical indices available in literature, (12a) the Mg$_2$
values from \citet{ben93} and (17) calcium Triplet from
\citet{cen03}.\\ 
References: (a)~\citet{smi00}, (b)~\citet{fal99},
(c)~\citet{bin85}, (d)~\citet{dac98}, (e)~\citet{den05},
(f)~\citet{weg03}, (g)~\citet{kob99}, (h)~\citet{dal91},
(i)~\citet{fab99}, (j)~\citet{ann07}, (k)~\citet{san07},
(l)~\citet{how05}, (m)~\citet{ram05}, (n)~\citet{tra98},
(o)~\citet{tra00}, (p)~\citet{oga08}, (q)~\citet{kee96},
(r)~\citet{lau89}, (s)~\citet{ber02}, (t)~\citet{oli95},
(u)~\citet{gar06}, (v)~\citet{lu93}.
%\end{table}